

Reviewer #1: I think it's a well written editorial and the authors listed several authoritative articles as well as advances in brain metastasis of breast cancer this year to support Chen YR's research. While emphasizing the importance of metastasis risk screening, the key points of treatment for patients with brain metastasis are also proposed.

Answer: Thank you for your evaluation and valuable comment.